Author’s response to reviews

Title: Critical Appraisal Skills of Family Physicians in Ontario, Canada

Authors:

Dr Marshall Godwin (godwinm@post.queensu.ca)  
Rachelle Seguin (seguinr@post.queensu.ca)

Version: 2 Date: 1 Nov 2003

Please accept the revised version of the manuscript "Critical Appraisal Skills of Family Physicians in Ontario, Canada". We would like to thank the reviewers for their helpful comments. Below is our response to the reviewers comments.

Reviewer: Chris Del Mar
We have added to the methods section the denominator of the College of Family Physicians database for Ontario physicians and explained that the random sample was taken from a computerized database capable of generating random samples.
We have tried to 'tighten up' some of the writing.

Reviewer: Roger Thomas
We explained why we had the respondent choose just two things to look for in the methods of articles.
We added the actual questions from the questionnaire.
All the typos were found and corrected.